Author's response to reviews

Title: Relationship between knee and ankle degeneration in a population of organ donors.

Authors:

Carol Muehleman (carol_muehleman@rush.edu)
Jun Li (jun_li@rush.edu)
Won Bae (wbae@ucsd.edu)
Koichi Masuda (kmasuda@ucsd.edu)

Version: 2 Date: 2 July 2010

Author's response to reviews:

no revisions